
diff -r 4cd75c619e06 -r cf3ca219c727 handouts/scala-ho.tex
--- a/handouts/scala-ho.tex	Tue Jan 12 02:49:26 2016 +0000
+++ b/handouts/scala-ho.tex	Mon Feb 22 22:09:31 2016 +0000
@@ -932,11 +932,17 @@
 \item \url{http://www.scala-lang.org/docu/files/ScalaByExample.pdf}
 \item \url{http://www.scala-lang.org/docu/files/ScalaTutorial.pdf}
 \item \url{https://www.youtube.com/user/ShadowofCatron}
+\item \url{http://docs.scala-lang.org/tutorials}
 \end{itemize}
 
 \noindent There is also a course at Coursera on Functional
 Programming Principles in Scala by Martin Odersky, the main
-developer of the Scala language.
+developer of the Scala language. And a document that explains
+Scala for Java programmers
+
+\begin{itemize}
+\item \small\url{http://docs.scala-lang.org/tutorials/scala-for-java-programmers.html}
+\end{itemize}
 
 While I am quite enthusiastic about Scala, I am also happy to
 admit that it has more than its fair share of faults. The




